

Temi di discussione

(Working Papers)

U.S. shale producers: a case of dynamic risk management

by Fabrizio Ferriani and Giovanni Veronese







Temi di discussione

(Working Papers)

U.S. shale producers: a case of dynamic risk management

by Fabrizio Ferriani and Giovanni Veronese

Number 1211 - March 2019

The papers published in the Temi di discussione series describe preliminary results and are made available to the public to encourage discussion and elicit comments.

The views expressed in the articles are those of the authors and do not involve the responsibility of the Bank.

Editorial Board: Federico Cingano, Marianna Riggi, Emanuele Ciani, Nicola Curci, Davide Delle Monache, Francesco Franceschi, Andrea Linarello, Juho Taneli Makinen, Luca Metelli, Valentina Michelangeli, Mario Pietrunti, Lucia Paola Maria Rizzica, Massimiliano Stacchini. *Editorial Assistants:* Alessandra Giammarco, Roberto Marano.

ISSN 1594-7939 (print)

ISSN 2281-3950 (online)

Printed by the Printing and Publishing Division of the Bank of Italy

U.S. SHALE PRODUCERS: A CASE OF DYNAMIC RISK MANAGEMENT?

by Fabrizio Ferriani^{*} and Giovanni Veronese^{*}

Abstract

Using more than a decade of firm-level data on U.S. oil producers' hedging portfolios, we document for the first time a strong positive link between net worth and hedging in the oil production sector. We exploit, as quasi-natural experiments, two similarly dramatic oil price slumps, in 2008 and in 2014-2015, and we show how a shock to net worth affects risk management practices among E&P firms differently depending on their initial financial positions. The link between net worth and hedging decisions holds in both episodes, but in the second oil slump we also find that leverage and credit constraints play a significant role in reducing hedging activity, a result that we attribute to the marked increase in leverage following the diffusion of shale technology. Finally, we test if collateral constraints also impinge the extensive margin of risk management. Though in this case the effect is less apparent, our results generally point to a more limited use of linear derivative contracts when a firm's net worth increases.

JEL Classification: D22, G00, G32.

Keywords: dynamic risk management, hedging, derivatives, shale revolution, oil price collapse.

Contents

1.	Introduction	5
2.	Literature review	10
3.	Data	13
	3.1 Data sources	14
	3.2 Net worth measures	15
	3.3 Derivative contracts	18
4.	Hedging and net worth	21
5.	Tackling endogeneity	23
	5.1 Instrumental variable estimates	24
	5.2 The role of leverage and credit constraints	27
6.	Optimal hedging strategy and net worth	31
7.	Conclusions	36
Re	eferences	38

^{*}Bank of Italy, Directorate General for Economics, Statistics and Research.

Hedging transactions expose us to risk of financial loss in some circumstances [...] Additionally, hedging transactions may expose us to cash margin requirements. Whiting 10-K

1 Introduction¹

Understanding corporate hedging strategies remains a fundamental challenge in modern corporate finance. Theoretical and empirical contributions have focused on the motivations and determinants of firms optimal hedging policies as well as on the effects of the hedging practices on firm value. However, firms' financial and operating policies are typically drawn jointly, so that establishing the causal effects of hedging or explaining hedging behavior can be a difficult task, hindered by endogeneity and reverse causality issues.

In this paper we investigate the hedging practices of U.S. companies in the oil exploration and production (E&P) sector. This setting is particularly amenable to analyze corporate behavior, as firms are exposed to a common risk factor (the oil price) and have access to a wide range of hedging instruments (Haushalter 2000; Haushalter et al. 2002; Jin and Jorion 2006). Oil producers hedge their production for a number of reasons including, but not limited to, commodity price risk management, lock in of cash flows to fund future investments, and loan covenants requiring minimum hedging amount. Haushalter (2000) offers a seminal contribution documenting substantial heterogeneity in hedging strategies in the E&P sector.

Our analysis covers the decade following the adoption of shale technologies. In this period the surge in production was phenomenal: U.S. crude oil production almost doubled from around 5 million barrels per day (mbd) in 2006 up to about 10.5 mbd in early 2018 (see Figure 1). The progress in hydraulic fracturing techniques has sustained increasing gains in the productivity and cost efficiency of E&P firms whose functioning has become more akin to a Hotelling producer. This evidence is presented in Bjørnland et al. (2017) where North Dakota firms using shale oil technology are found to be more flexible in allocating output intertemporally. Greater drilling responsiveness to oil prices is also documented in Newell and Prest (2017) and Anderson et al. (2018) who find from well-level data from Texas

¹The views expressed in this paper are those of the authors and do not necessarily reflect those of the Bank of Italy or the Eurosystem. The authors wish to thank Pietro Catte, Paolo Conteduca, Florian Heider, Taneli Mäkinen, Francesco Manaresi, Alfonso Rosolia, Enrico Sette, and seminar participants at Bank of Italy and NCB4 meeting for useful comments and suggestions on a previous version of this paper. All remaining errors are our own.



Figure 1 US oil production and leverage

The left plot displays the total US crude oil production and shale oil production measured in terms of million barrels per day (mbd); both series are from EIA. Shale-oil production includes hydraulically fractured production originated from EIA plays: Monterey, Austin Chalk, Granite Wash, Woodford, Marcellus Haynesville Niobrara-Codell, Wolfcamp, Bonespring, Spraberry, Bakken, Eagle Ford, and Yeso-Glorieta. The right plot displays median leverage defined as Total Debt/Assets for selected US E&P companies. Data are from Bloomberg, details on the firms included in the sample are available in Section 3.

that the drilling pattern, rather than oil production, is consistent with Hotelling's theory. By altering their supply response to forward prices at different horizons, the advent of the shale revolution may have translated also into changes in hedging strategies of oil companies (Domanski et al., 2015, Ferriani et al., 2018).

To finance the adoption of these new technologies oil producers have levered up substantially. As shown in the right plot of Figure 1, E&P firms accumulated debt to finance their operational growth in the most recent years; this resulted in a rapid increase of the U.S. oil and gas producing sector indebtedness, on the back of optimistic expectations regarding oil price developments (Domanski et al., 2015). This expansion occurred in a period of historically low interest rates with fairly stable oil prices (Azar, 2017). However, the buildup in leverage was not inconsequential for producers. According to Gilje et al. (2017) it materially affected firms' output and investment decisions and may have ultimately made the oil market more exposed to financial shocks (Dale, 2015).

Specifically, we explore how risk management by oil producers relates to increasing default risk and more tightening financial constraints. In the original literature pioneered by Froot et al. (1993) firms engage in hedging as a result of costly external financing; by hedging firms may limit the variability of internally generated cash flows and ultimately the amount of funds that are necessary to finance new investments. However, this theoretical prediction has been challenged by modern theories of dynamic risk management, claiming that limited or incomplete hedging characterize optimal risk management strategies for financially constrained firms (see Holmström and Tirole, 2000, Mello and Parsons, 2000). More recently Rampini and Viswanathan (2010) and Rampini and Viswanathan (2013) model the dynamic interplay between standard financing and risk management with collateral constraints. Their theoretical framework has been validated in recent studies documenting how financial constraints do actually limit firms' risk management (Rampini et al., 2014, Rampini et al., 2017).

We are the first to provide evidence of the important role of collateral in hedging decision in the oil producing sector, filling a gap in the literature on risk management Carter et al. (2017). Our empirical investigation relies on a new firm-level dataset of over 100 E&P U.S. oil producers, between 2006 and 2016. We hand-collected data from the companies' annual reports (10-k) on the notional amount of each hedging contract as well as on the different types of financial contracts used to hedge the oil annual production. With respect to other empirical studies investigating corporate risk management, our detailed dataset allows us to construct a more precise measure of the intensive margin of firms' hedging rather than a simply indicator signaling the use of derivative contracts (Guay and Kothari, 2003). In addition, thanks to the panel structure of our data, we are able to study the link between risk management and financial constraints not only by focusing on cross-sectional evidence but also exploiting sizable within-firm variation, a feature that we consider essential to study hedging dynamics in a sample of firms experiencing deep financial and technological changes. Finally, the additional information on derivative categories represents an imperative condition to extend the analysis to the optimal hedging strategies, an issue frequently disregarded in the literature because of data limitations (see Adam, 2009 or Mnasri et al., 2017b for some few exceptions).

The particular environment examined enables us to make progress on several dimensions, which are central to the literature on financial risk management. First of all, we focus on oil producers as in this context the functioning of the collateral channel is likely to be even more central in explaining risk management practices. Indeed, revenues for pure oil producer are more sensitive to oil price fluctuations compared to large and vertically integrated companies encompassing both upstream and downstream activities (Boyer and Filion, 2007 and Kumar and Rabinovitch, 2013). Moreover, the dynamic trade-off between firms' financing and risk management under collateral constraints is even amplified in this case as collateral pledged by oil producers takes the form of oil reserves, valued at market prices (see Office of the Comptroller of the Currency, 2018). We empirically document a strong positive link between net worth, representing our measure of financial constraints, and hedging in the oil producing sector. Our findings are robust to several checks. First, we consider different definition of net-worth, based on accounting as well as market based indicators to ensure that our results do not depend on a particular measure. Second, we do not limit to cross-sectional evidence but also look at the mere time-series dimension and test the impact on hedging by using net worth dynamics and a longitudinal setting. Third, we restrict the analysis to specific subsamples both time and firm-wide to account for periods of price declines and firms' defaults. Even more interesting, our results are confirmed also in an instrumental variable setting that we adopt to cope with possible endogeneity issues. We base our identification strategy on two different indicators: a measure of oil reserve revisions driven by exogenous price variation and an indicator of operational efficiency defined at well level. In both cases we find strong support for a positive causal link between a firm's net worth and its hedging ratio.

As a second contribution, we study how extreme market conditions in the oil market, leading to tighter financial conditions, influenced risk management in the US E&P sector. In the aggregate, hedging by oil producers recorded a sizable reduction following the two oil price collapses in 2008 and 2014-15. During these two episodes the net worth of oil producers represented in the right plot of Figure 2 was severely impacted. The net income/assets ratio was impaired especially after the 2014-15 slump, where years of accumulating leverage, combined with a more protracted price downturn, made the fragile financial conditions of shale oil producer even more dramatic and called for a more pronounced attention to their financial sustainability by lenders. We use these wide oil price fluctuations as a source of substantial



Figure 2 WTI price and net worth

The left plot displays the West Texas Intermediate (WTI) spot price with shaded area for the two significant oil price collapse in recent years; the series is from Datastream. The right plot displays median net worth defined as Net Income/Assets for selected US E&P companies; details on the firms included in the sample are available in Section 3.

variation in producers risk management strategies, through their effect on firms' net worth (see the left plot in Figure 2 for the dynamics of the WTI spot price in the last decade) to assess the intensity of the relation between hedging and net worth.² Interpreting these two dramatic oil price slumps as quasi-natural experiments, with a difference-in-difference approach we show how the shock to net worth differently affected the risk management practices among oil producers.

²We refer to Hamilton (2009), Kilian and Hicks (2012), Kilian (2014), Ellwanger et al. (2017), and Prest (2018) for some contributions explaining in detail the recent oil price dynamics and investigating the cause and consequences of the two oil slumps.

Finally, thanks to the wealth of information of our data set we are able to look into the particular contract chosen to hedge production and test whether net worth affects not only the hedging intensive margin but also the choice of the specific derivative category. This allows to further improve our understanding of hedging decisions by focusing also on the interlink between risk management motivations and contract characteristics, a theme that is not frequently investigated in the empirical research for scarcity of data. In this case we find a less evident relation linking net worth to firms' hedging strategies, though generally speaking our results points to a more limited use of linear derivative contracts when firms' net worth increases.

The rest of this paper is organized as follows. Section 2 reviews some of the theoretical and empirical contributions on firms' risk management, Section 3 describes the data set and the measures of net worth used to analyze firms' hedging strategies. Section 4 examines the main empirical results and discusses the identification strategy adopted to present evidence of a positive causal link between hedging and net worth. Section 6 supplements the analysis on hedging and net worth by analyzing how the latter impacts optimal risk management strategies. Finally Section 7 presents our conclusions.

2 Literature review

The literature has proposed a plenty of both theoretical and empirical contributions to explain risk management in non-financial firms. Departing from a classical Modigliani-Miller specification, we could generally claim that corporate hedging could be beneficial as long as it abates the impact of some market frictions including, but not limited to taxes, managerial risk aversion and compensation scheme, information asymmetries, costs of bankruptcy. In this paper, we explicitly focus on risk management as a result of costly external financing, a theme that was originally advanced in a seminal theoretical study by Froot et al. (1993). In absence of hedging, firms may be exposed to some variability in their internally generated cash flows which could potentially undermine their investment policies. To solve this problem firms can turn to external financing, but only up to some extent if "the marginal cost of funds goes up with the amount raised externally", for instance because of informational asymmetries or bankruptcy costs. On the contrary, by hedging, firms ensure they have sufficient internally generated funds when attractive investment opportunities arise so as to mitigate underinvestment. However, despite the appeal of their framework, little empirical evidence has been found for their model (Stulz, 1996). Since then, several studies have analyzed the interlink between financial constraints constraints and risk management, and how this may limit the firms' ability to hedge. Holmström and Tirole (2000) investigate the determinants of hedging modeling jointly liquidity management, risk management and capital structure. The authors suggest that credit constrained firms can preserve internal funds and deliberately choose an incomplete insurance against liquidity shocks to maximize the marginal return on investments in case the shock does actually materialize. Mello and Parsons (2000) characterize optimal hedging strategies for financially constrained firms. In their model the optimal hedge minimizes the variability in the marginal value of the firm's cash balances, by redistributing them across states and time. They show that a departure from a condition of perfect hedge can negatively affect the firm's expected costs of financing, its financial constraints, and the corporate value. More importantly for our analysis, their framework predicts limited or incomplete hedging for severely constrained firms, thus reconciling with the empirical evidence of higher hedging intensity among larger firms.

A more explicit challenge to the framework of Froot et al. (1993) is available in Rampini and Viswanathan (2010) and Rampini and Viswanathan (2013) who examine how collateral constraints alter the dynamic trade-off between external financing and risk management. They challenge the study of Froot et al. (1993) by relaxing two crucial assumptions. First, they dynamically consider investment and hedging as two competing firms' decisions; second, they assume that both activities absorb collateral to cover promises to pay. Therefore, it is precisely when a firm net worth falls and the marginal value of its investments is high, that it may choose to forgo hedging so to invest its limited resources. Their model is empirically validated in the case of airline companies hedging commodity price risk, see Rampini et al. (2014). Hedging falls among more financially constrained airlines and more so as they approach distress: airlines prefer to pledge the scarce collateral to finance investments rather than hedging fuel price risk. More recently, Rampini et al. (2017) unveil a similar trade-off between financing and risk management for US financial institutions when hedging interest rate and foreign exchange risk.

Though the theoretical framework in Rampini and Viswanathan (2010) and Rampini and Viswanathan (2013) has been so far tested in the context of commodity risk management for firms hedging an input price (i.e. airlines), analyzing risk management in E&P US oil companies during 2006-2016 permits to extend the empirical research in this field from several perspectives. First, as discussed in Section 3, in these firms the exposure of revenues and assets to a volatile risk factor is definitely higher and can exacerbate firms' financial con-

straints by severely impairing their balance sheet positions and the value of their collateral. Second, in small independent E&P firms the scope of business is generally limited to mere oil extraction, thus excluding oil refinement activities that can represent a source of natural hedging for large vertically integrated firms Mackay and Moeller (2007). Third, during the time-span of our sample the oil market was hit by two sharp price declines whose effects were also combined with significant spikes in market volatility, as measured by the CBOE crude oil volatility index. This allows us to study the dynamic trade-off between external financing and risk management during a period of market turbulence. Indeed, this is of particular interest as price uncertainty may curb corporate capital investment, with an effect that is even amplified when it is associated with limits to firms' ability to hedge (Doshi et al., 2017).

In the context of the oil sector, some contributions have recently added to the empirical research by analyzing risk management from several points of view. A test of the effectiveness of financial risk management is provided in Gilje and Taillard (2017), who show how hedging is effective in reducing the probability of financial distress and underinvestment risk, thereby affecting also firm value. Following the widening of the spread between US and Canadian oil price benchmarks, Canadian firms more exposed to basis risk are shown to reduce investment, record falling valuations, sell assets, and reduce debt. Focusing on US oil producers, Gilje (2016) studies how collateral based financing may distort investment decisions and trigger risk shifting behavior. He exploits an exogenous variation in leverage, induced by two episodes of marked oil price falls, to show that more restrictive covenants or a shorter duration of debt mitigate the risk-shifting, as proxied by the share of expenses in exploratory drilling over total investment expenditure. Gilje et al. (2017) use detailed well-level data to unveil a debt related investment distortion which emerges when producers, in the face of falling oil prices and a futures curve in contango, rush to complete wells and exploration in order to increase the value of their collateral, so to improve their credit standing. This acceleration in well completion is more pronounced ahead of regularly scheduled renegotiations with creditors. Similarly, Lehn and Zhu (2016) show that, during 2011-16, more leveraged companies faced with collapsing oil prices and declining investment opportunities, still ramped up production to meet debt repayments. Such a debt-driven investment distortions may even hinder the downward adjustment in oil production as oil prices fall, thus reducing the supply elasticity of the otherwise more price sensitive shale producers.³

³When collecting data on hedging we came across multiple evidence of how tied is the connection of oil producers with financing banks also with respect to risk management. An example is found in Whiting's 2015 10-K report: "Commodity derivative contracts held by the Company are with six counterparties, all of which are participants in Whiting's credit facility as well, and all of which have investment-grade ratings

Along with the decision of whether and how much to hedge (*intensive margin*) firms need to choose how to hedge (*extensive margin*). To do so, oil producers can customize their risk exposure by resorting to a menu of derivative instruments and strategies. Compared to the intensive margin of hedging, fewer studies have looked into how firms effectively hedge, probably as a result of the scarcity of firm-level information on derivatives portfolio. Limiting to commodity risk management, a seminal contribution by Tufano (1996) presents descriptive analysis on risk management strategies in gold mining firms, unveiling that the use of options is more frequent in large firms. Adam (2002) models firms' risk management in an inter-temporal setting to rationalize the firms' optimal decision in choosing between nonlinear and linear hedging strategies. His predictions are further explored in Adam (2009) who finds that options are prevalent in the gold mining industry, with financial constraints significantly increasing the likelihood of their adoption. Mnasri et al. (2017b) and Mnasri et al. (2017a) conduct an extensive analysis of the determinants and the value effect of nonlinear strategies for US oil producers. Finally, Croci et al. (2017) mainly focus on managerial preferences and risk incentives arising from executives' compensation schemes to describe the factors underpinning the optimal combination of derivative strategies. Our data set allows us to extend the analysis of derivative strategies in the oil market sector throughout a period including both new conditions in the financial markets (commodity financialization) and new producers relying on increasing leverage to lead the technological change in the oil production (shale revolution). In the following we will show that market dynamics have affected not only the extent of corporate hedging but also the mix between derivative contracts differencing in terms of payoff profile, cost structure, and complexity. In the last part of this study, we explicitly take into account collateral constraints for the choice of the hedging strategy by conditioning the extensive margin of linear derivatives to net worth measures as well as to a wide spectrum of firms' characteristics.

3 Data

In this Section we first describe the process of data collection based on hedging disclosure available in firms'annual report (10-K). Then, we provide some statistics on the derivative contracts employed to cover oil production and we define our measure of hedging activity. Finally, we present some possible measures of net worth that are used in the following

from Moody's and Standard & Poor. As of December 31, 2015, outstanding derivative contracts with JP Morgan Chase Bank, represented 76% of total crude oil volumes hedged.".

empirical analysis.

3.1 Data sources

To study the relationship between net worth and hedging we rely on a new firm-level dataset providing detailed information on E&P hedging strategies. More precisely, we hand-collected data for the period 2006-2016 on the notional amount of each hedging contract, as well as on the different type of contracts used to hedge the oil production. This information comes from annual company reports, available from the EDGAR website. First, we extract from Compustat the US companies classified with Standard Industrial Classification (SIC) code equal to 1311 (Crude petroleum & Natural gas) in the period 2006-2016. According to this criteria, there are 686 unique firms with primary activity in "Crude Petroleum and Natural Gas" exploration and production. Second, from this initial list we exclude those with either no publicly available 10-K on EDGAR, or with less than five years of reports.⁴ Third, we further filter out smaller reporting company that are not required to disclose information as their market risk is considered as negligible.⁵ This leaves us with 167 unique firms. Finally, we exclude those where risk management practices cannot be reclassified in terms of quantitative data as they are essentially not reported in tabular form in item "7A. Quantitative and Qualitative Disclosures about Market Risk". At the end of this filtering procedure we obtain an unbalanced sample of 102 unique firms observed over an 11 years time period. Some descriptive statistics on oil production, reserves and firm characteristics are shown in Table 1.

The firms in our final dataset account for approximately 30% of overall US oil production and are especially representative of small independent producers, including shale firms.⁶ This can be better grasped by comparing production obtained from our firm-level dataset with the one obtained from the US EIA statistics for shale regions. The evolution of production in the right plot of Figure 3 closely tracks the one of shale oil production presented in Figure 1, corroborating our choice of the sample to study firms which adopted shale technology. In the peak year of 2015, production reached 3.5 mbd when measured from our firm-level data, approximately 4.8 in the EIA statistics on shale production.

⁴This last choice is primarily motivated to ensure a minimum coverage of hedging practices over the period of analysis and make our dataset suitable to study within-firm evolution in risk management.

⁵We refer to the SEC website for further details on the definition of smaller reporting companies https: //www.sec.gov/smallbusiness/goingpublic/SRC.

⁶Major oil companies such as Chevron or Exxon Mobil, are not included in our sample, as they are generally classified in the 2911 SIC "Petroleum Refining".

Table 1Firms' summary statistics

	Ν	Mean	St.Dev.	25p	$50\mathrm{p}$	$75\mathrm{p}$
Oil production (Mb/d)	938	7.2	15.2	0.1	1.1	6.5
Oil reserves (MMb)	926	92.3	206.2	1.6	15.1	77.2
Production mix	941	44.9	31.2	16.4	42.3	70.0
Revenues (M\$)	944	1639.2	5319.9	44.0	296.1	1068.3
Property plant (M\$)	926	4136.0	8506.4	133.5	1025.6	3135.8
Capital expenditure (M\$)	962	976.6	1948.7	29.0	269.7	923.8
Employees	892	1033.5	2792.2	39.0	211.5	780.5

The table presents summary statistics for US E&P companies; data refer to firm-year observations. *Oil production* is crude oil produced measured in thousands of barrels per day; *oil reserves* is US proven developed and undeveloped reserves of crude oil held by the company at year-end, in millions of barrels; *production mix* is crude oil production as a percent of the company's total oil and gas production both measured in terms of barrel of oil equivalent; *revenues* is oil and gas revenues in US\$ millions; *property plant* is net value of property, land, and other physical capital in US\$ millions; *capital expenditure* is the amount spent on purchases of tangible fixed assets related to E&P activities in US\$ millions; *employees* are firm total employees.

The right plot of Figure 3 presents the median ratio between the value of net property and equipment (including oil and gas properties) over total assets. Oil and gas properties, which are largely made of proved reserves, represent a paramount component of total firm assets, with a median value larger than 80% across firm-years observations. The annual median value always fluctuates above 75%, with two sizable declines during the 2008/9 and the 2014/15 price slumps. In view of the limited size of non-oil related assets and revenues, this underscores the importance of the oil price in affecting producers' net worth and their borrowing capacity, which strongly reinforces the interplay between dynamic risk management and collateral availability, first described by Rampini et al. (2014). In their empirical application, airline companies need to hedge fuel oil costs, which account, on average, for between 20-30% of total operating costs, and firms pledge their assets (aircrafts) to borrow. In our case, oil producers need to hedge their output, and pledge oil reserves, both of which depend on the oil price.

3.2 Net worth measures

As the primary focus of this study is to test the impact of collateral constraints on hedging activity, we augment the data set to include information about firms' net worth. From this



Figure 3

Oil production and relevance of oil related assets

The left plot displays the total oil production in mbd of E&P firms included in the sample. The right plot shows the median ratio between net property and equipment over total assets; net property and equipment include oil and gas properties net of accumulated depreciation, depletion and amortization

perspective, we acknowledge that a correct identification of net worth represents a fundamental task for an empirical research analyzing the interplay between risk management and firms' financial constraints. Unfortunately, this purpose conflicts with the absence of an unambiguously accepted definition of net worth and in principle different measures of financial constraints could be accepted. For this reason, we partially follow previous studies in this field and use several balance sheet and market-based variables to construct a set of possible net worth indicators. This allows us to emphasize the robustness and consistency of our findings confirming that they are not subordinate to a specific measure. To this end, we combine firms' hedging data with a comprehensive list of financial and accounting variables retrieved from Bloomberg. We consider the following measures of net worth: net income/assets, the market value of equity (market capitalization), the ratio between the book value of equity and total assets, the book value of assets (size), and two market based measures implied by the Bloomberg Issuer Default Risk model, namely the 5 Year CDS and the 1 year probability of default.⁷ Descriptive statistics on net worth measures and firm leverage are reported in Table 2. Some net worth measures exhibit larger skewness and heterogeneity as concerns

	Ν	Mean	St.Dev.	25p	50p	75p
Net Income/Assets	945	-0.08	0.32	-0.09	0.01	0.06
Market cap.	881	6.60	2.14	5.22	6.72	8.15
Size	968	6.78	2.27	5.33	7.21	8.30
Equity/Assets	968	0.45	0.41	0.37	0.48	0.64
Bloomberg 5YR CDS	741	2.65	3.31	0.91	1.64	2.91
Bloomberg 1YR PD	760	1.20	3.14	0.01	0.11	0.81
Leverage	961	0.32	0.34	0.12	0.27	0.42

Table 2Net worth summary statistics and leverage

The table presents summary statistics for various new worth measures: *net income/assets* is net income divided by assets, *market capitalization* is log(number of shares*end of year price), *size* is log(assets), *equity/assets* is the book value of common equity divided by assets, *Bloomberg 5YR* CDS is 5 Year credit default swap spread for the company implied by the Bloomberg Issuer Default Risk model, *Bloomberg 1YR PD* is the probability of default of the issuer over the next 1 year calculated by the Bloomberg Issuer Default Risk model. *Leverage* is total debt divided by assets.

the frequency distribution, more so when considering net income/assets and the two market based indicators retrieved from Bloomberg. For these three measures the sample mean is quite far from the corresponding median and the distribution is quite dispersed. To a large extent this finding results from the two oil price slumps within our sample, which especially impacted on the market measures of net worth. In particular, the negative average value displayed for net income/asset reflects the severe net worth impairment experienced by some E&P companies during the two recent oil slumps reported also in the right plot of Figure 2.

⁷In some robustness tests we also try additional net worth measures including: cash dividends/assets, Bloomberg probability of default over the next 5 years, market value of equity/assets. The results are qualitatively similar and available upon request.

3.3 Derivative contracts

A benefit of examining the oil and gas industry is that disclosure of firms' derivative portfolios is remarkable. In particular, most firms provide information on each derivative contract, detailing the notional amount, the contract type, and maturity. Unfortunately this information is not presented in a standardized fashion, and data needs to be first hand collected and then reformatted. Starting from companies' 10-K report, we collect the information about the specific contracts used to hedge oil annual production.⁸

We then classify hedging instruments reported by companies in 8 distinct categories: futures/forward, swaps, collars, 3-way collars, swaption, call options, put options, and other derivatives including residual contracts. Table 3 displays the distribution frequency of the main class of financial instruments used to cover their 12-month ahead oil production. First of all, the table shows that about one third of the sample firms, in general the smallest producers in terms of size, are not engaged in risk management activities, with a remarkable peak around the 2008 oil slump. Moreover, hedging activity tends to be clustered into a limited number of derivative instruments, namely swaps, collars and three way collars.⁹ Finally, the table also shows an apparent variability over time in terms of the hedging strategy. This indicates that firms' dynamic risk management entails decisions not only with respect to the notional amount to be hedged but also in terms of optimal derivative contract.¹⁰

The second and third columns of Table 3 include hedging instruments that are classified as linear hedging contracts, i.e futures/forward and swaps. While the frequency of use of futures/forward contracts is marginal or nil for most of the years, swaps represent the most common hedging tool, around 50% over the full sample, peaking at 62% in 2013. The class of nonlinear contracts is very heterogeneous: it embraces multiple financial instruments with very different payoff structures. Among these, the importance of collars has remarkably shrunk over time, ranging from almost 50% at the start of the period to 23% in 2016.

⁸Being our focus on US oil producers, we only consider derivatives where the WTI is the price benchmark of the contract. These derivatives represent, for the firms in our sample, the most comprehensive category of contracts adopted to hedge oil production if not the totality itself.

⁹The analysis of how firms choose a specific derivative contract is out of the scope of this study, though Section 6 investigates the role of financial constraints as a determinant of general risk management strategies. Broadly speaking, firms may choose swaps to completely eliminate risk exposure, both upside and downside. On the other hand, collar and three way collar could be preferred to options because of their cost structure as well as because they maintain some degree of (probably desired) risk exposure (see Adam, 2009).

¹⁰Firms could in principle enter into derivative transactions to achieve a trading profit: however, from the majority of 10-K collected, companies explicitly indicate that their goal is to reduce the effect of oil price fluctuations as a leading motivations for engaging in derivatives transactions.

	No hedging	Linear co	contract Non-linear contract						
		Futures	Swap	Collar	3-way collar	Swaption	Call option	Put option	Other
2006	36.5	1.3	43.5	49.7	1.9	0.0	0.0	3.7	0.0
2007	36.0	0.0	46.1	43.5	4.5	0.0	0.6	0.1	5.2
2008	50.6	0.0	53.0	36.3	5.0	0.0	1.0	0.2	4.6
2009	44.3	0.0	54.2	37.2	4.5	0.0	0.6	0.7	2.8
2010	40.0	0.0	42.5	42.4	6.0	0.8	3.5	3.3	1.5
2011	32.7	1.5	44.5	34.7	9.1	0.3	2.1	2.8	5.1
2012	31.3	1.5	56.1	23.9	11.7	0.9	1.1	1.5	3.4
2013	28.4	0.0	62.0	18.9	10.6	0.7	1.5	2.4	3.8
2014	34.7	0.0	57.1	16.5	12.7	0.6	2.9	4.5	5.6
2015	38.0	0.0	43.2	22.0	17.2	0.0	5.9	6.7	5.0
2016	38.4	0.0	50.0	23.2	15.5	0.1	1.9	4.5	4.9
Total	37.1	0.4	50.8	30.1	9.5	0.4	2.0	2.8	3.9

Table 3Hedging choice and derivative contracts

The table displays the frequency of non-hedging firms and the frequency of use of different categories of hedging contracts. The table presents average values computed with respect to all firms reporting in a specific year; all values are in percentage. Category "Other" includes residual hedging instruments such as put spreads, enhanced swaps, fixed-price contracts which are not covered in specific categories.

Their decline has been almost entirely offset by the increase in the share of 3-way collars. Plain vanilla instruments, such as call and put options, have gained some use in the most recent years but remain marginal (5% on average in the full sample). Finally, residual financial instruments are used to cover almost 4% of the oil production, while the presence of swaptions is almost insignificant.

To study the hedging strategies of oil producers, we need to define a measure of production hedged. Rather than an indicator simply signaling the use of derivative contracts (Guay and Kothari, 2003), we construct an indicator measuring the intensive margin of firms' hedging activity as in Haushalter (2000). More precisely, we start from the notional amounts reported over all hedging contracts to cover the 12 month ahead oil production and define an annual ratio for the fraction of production hedged $HR12_i t$ as follows:



Figure 4 Sample hedging ratio

The graph displays the dynamics of the sample average hedging ratio which is defined as the ratio between total notional amounts reported over all hedging contracts to cover the 12 month ahead oil production and the oil production effectively achieved by the firm.

$$HR12_{it} = \frac{\sum_{j} Notional_{jit}}{Oil \ production_{it}}$$

where Notional_{jit} is the amount of hedged oil barrels for derivative contract j at time t as reported in the 10-K of firm i, while Oil production_it represents the annual oil production effectively achieved by the firm i. Figure displays the dynamics of the sample average hedging ratio $HR12_t$ for the E&P firms included in the sample. The average share of oil production hedged is 45% across firm-year observations, though with some skewness and variability in the whole sample. Average $HR12_t$ recorded two marked reductions following the oil price slumps (2008 and 2014/15), whereas hedging increased between these two episodes fostered by rising production and fairly stable oil prices. This finding does not, per se, allow us to draw any final conclusion on the causal relation between firms' net worth (closely determined by oil prices) and risk management; yet it presents some preliminary empirical evidence of an inverse relationship between the intensity of hedging and oil prices.

4 Hedging and net worth

The key assumption in Rampini and Viswanathan (2010) and Rampini and Viswanathan (2013) is that collateral constraints apply to both external financing and hedging activity. Hence, managers face a trade-off: they can either engage in hedging motivated by risk aversion concerns or they can preserve resources to increase their investments, more so when firms' net worth is low and marginal productivity of capital is higher. A vivid example of this trade-off is provided by the dramatic decision of Continental Resource to settle, in the aftermath of the 2014 oil price slump, all of its derivative contracts prior to expiration. In doing so the company remained completely unhedged and exposed to volatility in crude oil prices but at the same time was able to cash in the gains on its existing derivatives and fund its drilling programs.

In this Section we investigate if and how different measures of firms' net worth are positively related with firms' hedging activity. Our specification is based on the following baseline equation:

$$HR12_{i,t} = \alpha + \beta NW_{i,t} + \epsilon_{i,t} \tag{1}$$

where for each firm *i*, HR12 represents the hedging ratio (production hedged as a ratio of total production) for the 12 month ahead and NW is one of the net worth measures defined in Section 3.2; *t* denotes time measured in years. In general, if the crowding out effect between hedging and collateralized financing is active, we would expect β to be positive and statistically significant. Clearly, a negative relationship would hold for our market-based measures of default risk, such as *Bloomberg 5YR CDS* and *Bloomberg 1YR PD*, which are inversely related to the firms' net worth.

We present in Table 4 the results for the empirical specifications of interest. All models include firm fixed effects to account for the effect of time-invariant unobserved firm characteristics (e.g. managerial risk aversion, risk management skills) on hedging policies.¹¹ The first column of Table 4 presents fixed effect panel estimates and provides some strong evidence of a direct relationship between net worth and the amount of hedging: all coefficients display the expected sign at the 1% significance level except for size that is significant only at the 5% level. The magnitude of the effect varies across net worth measures (shown row-wise), but is economically relevant. For example, and with reference to the first column of Table

¹¹As a robustness check we also estimate a weighted least square specification and a tobit model to take into account the mass of firms signaling no hedging activity. The results are qualitatively similar and available upon request.

	(1) Panel est.	(2) Panel est. year FE	(3) First difference	(4) No crisis years	(5) No default. firms	(6) Only default. firms
Net Income/Assets	$\begin{array}{c} 0.231^{***} \\ (0.06) \end{array}$	0.163^{**} (0.07)	0.139^{**} (0.07)	$\begin{array}{c} 0.375^{***} \\ (0.14) \end{array}$	$\begin{array}{c} 0.188^{***} \\ (0.05) \end{array}$	$0.407^{**} \\ (0.17)$
Market cap.	0.086^{***} (0.02)	0.043^{*} (0.02)	$\begin{array}{c} 0.114^{***} \\ (0.03) \end{array}$	$\begin{array}{c} 0.083^{***} \\ (0.03) \end{array}$	0.096^{***} (0.02)	0.034 (0.07)
Equity/Assets	0.190^{***} (0.06)	0.139^{**} (0.05)	0.094^{**} (0.04)	$\begin{array}{c} 0.239 \\ (0.15) \end{array}$	0.245^{***} (0.09)	0.130^{**} (0.05)
Size	0.080^{**} (0.04)	0.073^{*} (0.04)	0.204^{**} (0.09)	$\begin{array}{c} 0.112^{**} \\ (0.05) \end{array}$	$0.057 \\ (0.04)$	0.167^{**} (0.07)
Bloomb. 5YR CDS	-0.019^{***} (0.00)	-0.013^{**} (0.01)	-0.012^{**} (0.01)	-0.024^{**} (0.01)	-0.017^{***} (0.00)	-0.049^{*} (0.03)
Bloomb. 1YR PD	-2.443^{***} (0.84)	-1.566^{*} (0.82)	-1.375^{*} (0.78)	-3.194^{**} (1.42)	-1.908^{**} (0.78)	-5.078^{***} (1.72)

Table 4Net worth and hedging

The table presents regression coefficients for the equation relating firms' hedging ratio and net worth. All models include firm fixed effects. Robust standard error in parentheses. Dependent variable is 12-month ahead hedging ratio; HR12 is equal to zero for non-hedging firms. *, **, and *** denote significance at, respectively, the 10%, 5% and 1% level. Net Income/Assets is net income divided by assets, Market Cap. is log(number of shares*end of year price), Size is log(assets), Equity/Assets is the book value of common equity divided by assets, Bloomberg 5YR CDS is 5 Year credit default swap spread for the company implied by the Bloomberg Issuer Default Risk model, Bloomberg 1YR PD is the probability of default of the issuer over the next 1 year calculated by the Bloomberg Issuer Default Risk model. Column 4 displays estimates excluding the years of oil price collapse (2008, 2014, 2015). Defaulted firms include: Berry Petroleum, Emerald Oil, Energy XXI, Escalera Resources, Goodrich Petroleum, Magnum Hunter Resources, Miller Energy Resources, Osage Exploration & Development, Postrock Energy, Sabine Oil & Gas, Sandridge Energy, Stone Energy, Ultra Petroleum.

4, a one standard deviation increase in net worth implies an increase in the hedging ratio varying on the basis of the net worth measure and ranging between about 6% (Bloomberg 5 YR CDS) and more than 18% (Market capitalization)¹². To fully exploit the within-firm variation in hedging behavior and rule out serial correlation effects we consider two different specifications: first, by including annual fixed effects (column 2); second, by considering the

¹²Similar, and sometimes even larger effects, are found for all the empirical specifications displayed in Table 4; results are available upon request.

baseline specification estimated in first differences (column 3). Albeit the statistical and economical significance of our results is somehow weakened, the estimated coefficients still point to a substantial effect across all net-worth measures.

To check whether our results are driven by the years with falling prices, affecting firms' net worth and forcing managers to reduce hedging and prefer borrowing when collateral is scarce, we also re-estimate the same specification by selecting specific sub-periods. When we exclude the oil price slump years (2008, 2014, 2015, model in column 4), we actually find that the opposite is true, as the standardized effect of net worth on hedging is generally even larger. In the same spirit, in the last two columns of Table 4 we test the model by splitting the sample between firms which never defaulted in our sample (column 5) and those which have defaulted (column 6).¹³ This is to check if the results reported in columns (2)-(4) are driven by distressed firms included in the sample. Indeed, we would expect defaulted firms to have reduced more intensely their hedging, as collateral constraints become even more binding in this case. The estimates in Columns 5 and 6 seem to rule out also this possibility, though the statistical significance of estimates referred to defaulted firms is not always very strong, which is not surprising in light of the considerably smaller number of observations used.

In general the findings presented in this Section provide, in the context of US oil producers, a strong empirical validation of the link between net-worth and hedging, emphasized by modern dynamic risk-management theories. This result is remarkably robust both across the range of net-worth measures considered and various model specifications.

5 Tackling endogeneity

In the previous Section we showed that less financially constrained firms engage more in risk management activities. Though this relation seems robust across multiple model specifications and definitions of net worth, omitted variable bias and simultaneity may represent a potential concern. In this Section we address this issue. First, we present instrumental variable (IV) estimates using an identification strategy that exploits E&P firms' main source of net worth, namely oil reserves, as well as a measure of firms' operational efficiency. Second, we employ the oil price declines in 2008 and 2014-15 as quasi-natural experiments to show

¹³We consider as defaulted the following firms filing for bankruptcy: Berry Petroleum, Emerald Oil, Energy XXI, Escalera Resources, Goodrich Petroleum, Magnum Hunter Resources, Miller Energy Resources, Osage Exploration & Development, Postrock Energy, Sabine Oil & Gas, Sandridge Energy, Stone Energy, Ultra Petroleum.

how companies remarkably reduced their hedging activity, as they became more financially constrained when hit by these two severe shocks to revenues.

5.1 Instrumental variable estimates

For the IV exercise we focus on net income/assets among the possible measures of net worth. This indicator is a flow variable which successfully captures net worth variations as a consequence of oil price dynamics (see Figure 2). Moreover, to rule out spurious results because of variations in the operating scale of the company, this measure is also standardized by total assets. We consider two possible instruments for net worth. First, we rely on an identification strategy that uses changes in reported oil reserves as a source of idiosyncratic variation in the firms' net worth.¹⁴ Oil reserves account for a substantial fraction of E&P companies' net worth and represent the principal asset component in their balance sheets, as depicted in Figure 3. Moreover, oil reserves define the common source of collateral in the context of the so called "reserve base lending", i.e. the standard financing process of E&P firms where the amount of money granted is proportional to the extent of proven oil reserves (Domanski et al., 2015, Azar, 2017). To construct our first instrument, we exploit a unique feature of companies annual reports, which provide detailed information of the factors driving changes in the amount of proved oil reserves: acquisitions, sales, extensions and new discoveries, production, and revaluation. This allows to discriminate between changes in net-worth due to managerial decisions (e.g. to drill more to expand the reserve base) and hence tightly related to the hedging decision, from those "sufficiently" exogenous to the firms decision. To this end, we focus on the variation in reserves due to the revaluation component and compare this item to the total amount of reserves available to the firm. Based on SEC definition "Proved O & G reserves are quantities of petroleum that, by analyzing geological and engineering data, are reasonably certain to be commercially recoverable from a given date forward from known reservoirs and under current economic conditions, operating methods, and government regulations". In other words, the reserve revision item accounts for variation in the total amount of reserves due to factors such as updated reservoir analysis, well performance, and especially variation in oil prices; indeed, we find extensive anecdotal evidence in the companies' 10-K that changing economic conditions are the main driver of annual reserve revision.¹⁵. Hence, our first instrument for net worth is defined as:

¹⁴A similar strategy is used in Gilje ($\overline{2016}$).

¹⁵For the sake of brevity we report only some examples included in the following 10-K: Carrizo (2015): Crude oil, condensate and natural gas liquids revisions of previous estimates are primarily attributable to the following: 2015 negative price revisions as a result of the significant decrease in the oil price used to calculate

$Instrument \ 1 = \frac{Reserves \ Revision}{Reserves}$

where *Reserves Revision* quantifies the variations in reserves due to exogenous factors and *Reserves* represent the amount of company's oil reserves (both variables are measured in physical oil barrels). Our identification strategy is based on the assumption that variation in oil reserves, net of the production component and other recomposition effects driven either by sales or purchases of properties, should affect the intensive margin of hedging through their impact on the firm's net worth. In fact, oil price dynamics exogenously determines a revaluation of reserves which is unrelated to managerial decisions potentially affecting other firms' characteristics, such as risk management practices. Nevertheless, in order to mitigate residual concerns on endogeneity, in the IV estimation we also control for oil market conditions and firms' oil production.

Second, as an alternative instrument, we consider a firm-level efficiency indicator of the exploration activity. In principle, we expect more productive firms to be also the ones with higher net worth. Our identification strategy hinges on the assumption that drilling efficiency, arguably a dimension of productivity relating to the physical and geological characteristics of the oil fields being drilled, while affecting firms' net worth should not be linked to the financial management decision.¹⁶ To this end, we use an indicator of well efficiency defined as follows:

$$Instrument \ 2 \equiv Well \ efficiency = \frac{Gross \ Productive \ Wells}{Employees}$$

where *Gross Productive Wells* represents the number of wells producing, or capable of producing, oil at year-end including also the wells in which the company has a full or partial interest, while *Employees* is the number of company's workers. To deal with potential endogeneity, in this case we control for factors related to firms' drilling activity, such as capital

our proved oil reserves estimates of 11,194 MBbls, partially offset by positive performance revisions of 4,904 MBbls.; Pioneer (2012): at December 31, 2012, revisions of previous estimates are comprised of 82 MMBOE of negative price revisions and 27 MMBOE of negative revisions due to updated performance profiles and cost estimates; Continental resources (2015): Commodity prices decreased significantly in 2015 [...] These decreases shortened the economic lives of certain producing properties and caused certain exploration and development projects to become uneconomic which had an adverse impact on our proved reserve estimates, resulting in downward reserve revisions of 185 MMBls [...]. We may experience additional downward reserve revisions as a result of prices in 2016 if the currently depressed price environment for crude oil and natural gas persists or worsens.

¹⁶In Rampini et al. (2014) the authors instrument net worth with changes in productivity as proxied by firms' operating income. Mansell et al. (2012) show that the average cost per well drilled is lower among larger E&P firms.

		First stage equation					
		IV1	IV2	IV1-IV2			
Instrument 1		0.292**		0.149**			
Instrument 2		(0.11)	0.015***	(0.06) 0.020^{***}			
			(0.00)	(0.01)			
F statistics		8.831	20.828	11.682			
		Seco	nd stage eq	luation			
	Panel	IV1	IV2	IV1-IV2			
Net Income/Assets	$ \begin{array}{c} 0.231^{***} \\ (0.06) \end{array} $	$\begin{array}{c} 0.369^{**} \\ (0.19) \end{array}$	$\begin{array}{c} 0.437^{***} \\ (0.16) \end{array}$	$\begin{array}{c} 0.494^{***} \\ (0.17) \end{array}$			
Hansen J p-value	_	_	_	0.868			

Table 5IV regression - instrumenting net worth with reserves and well efficiency

IV regression with instrumented net worth measures and firm fixed effects; robust standard errors in parentheses. Dependent variable is net income / assets in the upper panel (first stage equation) and HR_12 in the lower panel (second stage equation). In Column IV1, net worth is instrumented via Instrument 1 (*Reserve Revision/Reserves*), while in IV2 net worth is instrumented via Instrument 2 (*Well efficiency*); in Column IV1-IV2 both instruments are jointly included in the specification. Controls in IV1 regression include the annual average of the CBOE Crude Oil Volatility Index (CBOE oil vix), the annual average calendar strip on WTI, and the firm's crude oil production. Controls in IV2 regression include firm-level measures of costs incurred for E&P activities, capex expenditures, business efficiency per employee (EBIT/employee), drilling success rate, and crude oil production. All the controls in IV1 and IV2 regressions are included in the joint estimation IV1-IV2. *, **, and *** denote significance at, respectively, the 10%, 5% and 1% level.

expenditures and success rate in the development of new oil wells.

Our IV estimates (2SLS) are reported in Table 5. The upper panel presents the results from the first stage regression: all the three specifications display a positive correlation between net worth and our proposed instruments, which still holds when they are jointly considered (column IV1-IV2). Also the F-statistics from the reduced form equations points to an adequate relevance of revision in oil reserves and drilling efficiency as instruments for net worth.¹⁷ In the lower panel of Table 5 we provide the second stage estimates of the link between risk management and instrumented net worth; for ease of comparison we also

¹⁷Admittedly, the F-statistics in the IV1 specification is marginally lower than the common applied threshold of 10. Though this could potentially mask a problem of weak identification, we note that the value of our F-statistic is also remarkably impacted by the adoption of clustered standard errors.

report, in the first column, the estimates obtained earlier when net worth was proxied by net income/assets in Table 4. As anticipated, all specifications are augmented by a set of regressors accounting for oil market conditions, firm investments, crude oil production, and drilling activity. The evidence in Table 5 confirms the theoretical prediction of a positive causal relation linking net worth to hedging. All the IV estimates are qualitatively comparable with those previously reported in Table 4. However, the magnitude of the effect estimated by IV is generally larger than the one previously obtained; this results points to a possible measurement error attenuation respect to the panel estimate.¹⁸ Finally, for the combined specification IV1-IV2 we also report the Hansen J-statistics for the test of overidentifying restrictions, which does not reject the null hypothesis of validity of our instruments. Overall, the findings in Table 5 seem to validate our identification assumptions based on variations to operational efficiency and the value of oil reserves, as two factors affecting risk management practices through their impact on firms' net worth.

5.2 The role of leverage and credit constraints

In the following we shed further light on the relationship between net worth and risk management. To this end, we exploit the two oil price collapses in our sample, i.e. the 2008 and 2014-15 oil price slumps, as quasi-natural experiments. These oil price declines were an exogenous and dramatic shock to oil companies' net worth which markedly impaired their borrowing capacity. If higher net worth is indeed a key factor driving the interplay between hedging and collateralized external financing, we would expect a decrease in hedging for firms more deeply affected by the commodity shocks. As pointed out by Mello and Parsons (2000), every hedging strategy comes packaged with a borrowing strategy. Suggestive evidence for a tight link between between credit and hedging decisions comes from the 2015 10-K filing from Whiting: the credit agreement contains restrictive covenants that may limit our ability to, among other things [...] enter into hedging contracts, incur liens and engage in certain other transactions without the prior consent of our lenders.

We resort to a difference-in-differences strategy (DID), where we separately test the effect of the two price declines by symmetrically splitting our data range in 2011.¹⁹ This

¹⁸At least for IV1 where net worth is instrumented via exogenous variation in reserves, we stress that a full comparison cannot be achieved as data on oil reserves are available for the whole sample starting from 2010.

¹⁹The main results discussed in this Section are qualitatively similar with different choices of the splitting year (e.g. 2010 or 2012) as well as if we exclude observations during the oil slumps years (2008, 2014, 2015). The full set of estimates is available upon request.

choice allows to take into account two important issues. First of all, firms may have been differently affected by the two oil collapses, so assuming a dynamic treatment threshold is fundamental for dealing with the time-varying classification of firms (treatment vs control) in the two events and with sample attrition because of bankruptcies. Second, starting from 2010-2011 E&P companies have been facing not only a technological development with shale oil boom but also a profound transformation of their financial structure as reported in Figure 1. Therefore, by splitting the sample in two periods allows to appreciate the impact of the buildup in debt observed during the shale revolution.

We proceed by assuming a within-event matching, and we construct treatment and control groups on the basis of the companies net worth in the year prior to the crisis. For example, in the DID regression for the 2008 oil slump, a firm is considered as treated when its net worth measured by net income/assets is below the median value of the sample net worth in 2007. A similar strategy is adopted for the 2014-15 case, using the median net worth in 2013.²⁰ The effect of a commodity shock on the hedging ratio of firms is evaluated through a DID setting according to the following regression form:

$$HR_{it} = \alpha + \beta_1 Post + \beta_2 Treatment_i + \beta_3 Treatment_i \times Post + \epsilon_{it}$$
⁽²⁾

where *Post* is a dummy variable equal to 1 starting since the occurrence of the commodity shock (2008-2011 or 2014-2016), while *Treatment* is another dummy variable that is equal to 1 for firms classified as treated, i.e. with net worth below the sample median. In the DID regression the coefficient of interest is the one of the interaction term, β_3 . This coefficient measures the difference between pre-shock and post-shock hedging behavior for treated firms relative to firms whose net worth is less deeply affected by the decline in oil prices. Table 6 displays the DID estimates for the two episodes, with the 2008 results in the top panel and those for 2014-15 in the bottom panel, respectively. Bearing in mind that oil-related assets account for the lion's share in E&P companies net worth we would expect, in line with theoretical predictions, the interaction coefficients (β_3) to exhibit a negative and statistically significant sign. This is indeed what emerges from Table 6, confirming the causal link between net worth and hedging. In both episodes, the magnitude of the effect is also economically relevant and comparable to the one reported in Table 4.

The significance of this additional reduction in hedging for firms with lower net worth is robust to the inclusion of additional variables accounting for firm leverage and short-

²⁰In both episodes we also tried different thresholds for net worth, e.g. percentiles ranging in the interval 50-75. The estimates are qualitatively similar to the median case and are available upon request.

Table 6Difference in difference estimates

		2008 oil	slump	
	1	2	3	4
Year>=2008× Treated NW	-0.264^{**} (0.123)	-0.273^{**} (0.127)	-0.275^{*} (0.144)	-0.256^{**} (0.116)
Short term obligations		$0.150 \\ (0.201)$		
Available Credit lines			-0.011 (0.048)	
Leverage				-0.251 (0.543)
		2014-15 o	il slump	
	1	2	3	4
Year>=2014× Treated NW	-0.186^{**} (0.093)	-0.176^{**} (0.088)	-0.174^{*} (0.099)	-0.182^{**} (0.091)
Short term obligations		-0.113^{***} (0.008)		
Available Credit lines			0.066^{**} (0.027)	
Leverage				-0.130^{**} (0.055)

Difference-in-differences estimates with firm fixed-effects. Robust standard error in parentheses. Dependent variable is 12-month ahead hedging ratio. A firm is considered as treated when its net worth measured by net income/assets is below the median value in the year prior to the 2008 or 2014-15 oil price shocks, respectively. *Short term obligations* represent all debt and payments due within one year, *available credit lines* represent the unencumbered fraction of credit lines granted to the firm, *leverage* is measured as total debt/assets *, **, and *** denote significance at, respectively, the 10%, 5% and 1% level.

term debt constraints. Controlling for this credit dimension is not only relevant to explain the transformation in the financial structure of oil firms, but it is also crucial in order to account for other factors affecting firms' hedging decisions. To this end we consider separately, to avoid collinearity issues, three measures of financial constraints: short-term obligations which measure the amount of obligations expiring within one year, available credit lines that provide information on the fraction of unencumbered bank borrowings, and leverage defined as total debt/assets. The two last variables control, respectively, for shortterm commitments that could curb additional risk taking and for unused debt capacity which could enhance firm liquidity. When we augment the baseline DID regression with the proxy of firms' indebtedness we continue to find a significant effect of net worth on hedging. However, comparing the estimates across the two panels of Table 6, we find that the additional controls for "credit constraints" display a statistically significant coefficient only in the second episode, and at the same time also the magnitude for the interaction term is somewhat dampened. One potential explanation for this result can be sought again via the graphical insight presented in Figure 1 and Figure 2. The second price fall stroke the oil industry after a period of increasing debt accumulation driving the expansion of shale oil production (Domanski et al., 2015). As an example, the leverage ratio in 2013 was about 8 p.p. higher than the 2008 value, peaking to a level 16 p.p. higher in 2015 after the fall in output prices. The surge in leverage and the differences in financial conditions become more apparent in the second half of the sample span, making firms more financially constrained and close to distress which could explain the different outcome for the two oil shocks. In other words, starting from 2010 the burden of the debt component in the "collateral call" has become larger and showed up also directly in the leverage variables, while earlier the same effect was not apparent.²¹

To check the validity of our quasi-natural experiment and validate the causal interpretation of results, we end this Section by presenting the results of a placebo test. We restrict the sample to the period 2009-2013 and create a placebo event in 2011 to examine if treatment and control firms engage differently in risk management also in time periods where no relevant oil price decline occurs. In this case a firm is considered as treated if its net worth measured by net income/assets is below the median value of the sample net worth in 2010. Results for the placebo test are displayed in Table 7. The interaction terms are always not statistically significant and also considerably smaller in terms of magnitude compared to the estimates reported in Table 6, pointing to no relevant differences in risk management activities among treated and control companies conditional to their level of net worth.

²¹Smith and Stulz (1985) suggest that direct and indirect cost of bankruptcy should be a key determinant of firms' hedging policies. We do not find evidence to support their prediction with our estimates being more in line with the collateral constraint theory affecting risk management activity; interestingly, our findings still hold even if we exclude distressed firms from the sample.

Table 7 Placebo tests

		Placebo test					
	1	2	3	4			
Year>= $2010 \times$ Treated placebo	-0.001	0.003	-0.001	0.063			
	(0.084)	(0.086)	(0.085)	(0.099)			
Leverage		0.354					
5		(0.267)					
Short term obligations			-0.071				
			(0.378)				
Available Credit lines				0.050			
				(0.048)			

Difference-in-differences estimates with firm fixed-effects. Robust standard error in parentheses. Dependent variable is 12-month ahead hedging ratio. A firm is considered as treated when its net worth measured by net income/assets is below the median value in 2010. Estimation sample is 2009-2013 for the placebo test. *, **, and *** denote significance at, respectively, the 10%, 5% and 1% level.

6 Optimal hedging strategy and net worth

The previous Sections provided extensive evidence of net worth as a major determinant of firms' hedging decision, an effect amplified when the value of firms' collateral is impaired by severe oil shocks. In this last Section we examine more in detail how optimal hedging strategy and the extensive margin of risk management activities interact with firms' net worth, an aspect often neglected in previous studies on commodity producers.²² Nevertheless, as shown in Table 3, preferences of firms between linear and nonlinear strategies as well as their choice in terms of specific derivative contracts have evolved over time.

Financial derivatives adopted by E&P firms to hedge oil production differ both with respect to their final payoff structure and in costs, as well as in how they can affect the firms' collateral needs. To the best of our knowledge the only papers specifically devoted to the analysis of the optimal hedging mix in the oil industry are the one by Mnasri et al. (2017b) and Croci et al. (2017). We depart from their approach from several perspectives. First, as previously discussed, we fit our analysis in the framework of dynamic risk management theories, where collateral constraints impinge on the firm ability to engage in derivatives

 $^{^{22}}$ Several authors presented a theoretical framework for the choice of the hedging strategy see Smith and Stulz (1985), Adler and Detemple (1988), Froot et al. (1993), Brown and Toft (2002), and Adam (2002) among many others.

trading. Hence, we explicitly condition the choice of hedging (the extensive margin) to net worth, as well as to financial constraints. Second, we are the first to examine oil producers hedging strategies in the aftermath of the shale technology. As discussed in Section 1 this transformation not only altered the production from the technological point of view, but also the firms' financial structure via increasing leverage. Moreover, this financial transformation occurred during a period where a severe shock to oil prices particularly stroke small independent E&P producers whose financial hedging could not be replaced by other form of risk management (Kumar and Rabinovitch, 2013).

We focus on the extent of linear hedging measured as oil production covered via linear contracts over total oil production hedged. In this way, we construct an indicator of hedging strategies which is normalized to one, so that the natural complement to linear hedging includes all the remaining oil production hedged via collars, three-way-collars, put options, call options, swaption and other residual contracts. We prefer investigating the extent of linear hedging instead of considering specific nonlinear strategies in the spirit of Adam (2009) for several reasons. First, linear contracts represent a definitely more homogeneous category and their analysis does not require to distinguish among nonlinear contracts with very different payoffs and underpinning strategies.²³ Second, the heterogeneity among nonlinear contracts does not always support a clear-cut identification of the main determinants of the optimal hedging strategy, and the available empirical findings are sometimes inconclusive to this purpose, see Adam (2009) or Croci et al. (2017). Finally, adopting an empirical specification where the dependent variable is split across multiple distinct categories could be problematic for some types of nonlinear derivatives whose use is almost minimal in specific sample years (see Table 3).

We estimate a Heckman model to control for sample selection in the hedging decision.²⁴ In the first stage, the dependent variable is a binary dummy equal to 1 when the company's 10-K reports derivative contracts in place to hedge oil production. In the second stage, we evaluate the preference for linear hedging by measuring the extent of total notional hedged through linear contracts over the total notional amount of the hedging portfolio; in both stages we include one of our measures of net worth and we control for several variables which have been identified to play a role in shaping risk management strategies. The estimates

 $^{^{23}}$ For example, a costly short put with no upside cap is typically adopted for insurance purposes, while a three-way-collar may even generate a profit thanks to an additional sold put, but it has no downside protection for very low price levels.

²⁴For the sake of brevity the analysis in the previous sections ignored the selection bias for hedging as our main results are not qualitatively affected. Indeed, the significance of the inverse Mill's ratio in the Heckman model is not uniform across net worth measures.

are presented in Table 8, with the selection equation in the bottom panel and the outcome equation in the top panel.

Our selection equation results confirm the evidence on the role of net worth presented in Section 4, also when several controls are included in the specification, with the only exception of the 5 year CDS measure which is still negative but not statistically significant. On the other hand, collateral constraints seem to influence only marginally the decision about the optimal amount of linear contracts. In the outcome equation, a higher net worth is associated with a smaller share of linear hedging, though this result is not uniform and depends on each specific measure. In principle, we acknowledge that a more precise analysis of this result would require detailed information on the extent of collateral absorption by different hedging contracts. Unfortunately, this kind of data is not easy to recover as most of risk management is carried out via OTC contracts with very limited public disclosure. Nevertheless, we believe that some speculations can be made to support these findings. A first one relates to the organizational development of firms, i.e. to the fact that the risk management function could be less developed and skilled in firms with lower net worth, thus explaining their preference for more naive derivative strategies. Alternatively, and probably more interesting in the context of dynamic risk management, linear hedging though being subject to margin calls, does not necessarily require an up-front premium and could be preferred by firms with lower net worth. Conversely, a higher net worth and less tighten financial constraints could explain the use of more complex and expensive nonlinear strategies, which however preserve the upside potential (see also Adam, 2009 on this point).²⁵

The impact of the oil price on the decision of whether to hedge is never significant; on the contrary, and not surprisingly, increasing levels of oil production positively affects the probability to enter a derivative contract. Oil price is also found to be strongly positively correlated with the extent of linear hedging. Linear contracts such as swap and forwards allows firms to hedge their production but they do not represent a profitable strategy when oil prices increase because of a cap to the upside potential with respect to nonlinear strategies, a result already documented in Adam (2002) and Adam (2009) for gold mining firms.

Production risk, defined as firm-specific coefficient of variation of oil production, is positively associated with the extent of linear contracts, though this findings is not always confirmed across specifications. Admittedly Brown and Toft (2002) and Gay et al. (2002)

²⁵Some anecdotal evidence of this assumption can be recovered from the increasing adoption of 3-way collars starting from 2011 with producers being increasingly confident on the historically high oil prices. Hence, if the use of 3-way collars was originally conceived to realize a better upside exposure, it proved to be a very costly and ineffective hedging strategy after the oil price collapse in 2014-2015.

	Income/	Market	Equity/			Default
	Asset	Cap.	Asset	Size	CDS	1 year
			Outcome	e equation		
Net worth	0.048	-0.038***	0.004	-0.037**	-0.005	0.016**
	(0.072)	(0.012)	(0.170)	(0.015)	(0.007)	(0.006)
Oil price	0.318***	0.350***	0.288***	0.284***	0.381***	0.399***
-	(0.101)	(0.097)	(0.096)	(0.096)	(0.109)	(0.104)
Production uncert.	0.128^{*}	0.088	0.128^{*}	0.080	0.299***	0.284^{***}
	(0.071)	(0.073)	(0.074)	(0.073)	(0.087)	(0.086)
Investments	-0.132***	-0.120***	-0.134^{***}	-0.142^{***}	-0.223**	-0.216**
	(0.045)	(0.042)	(0.046)	(0.048)	(0.094)	(0.087)
Profit diversif.	-0.308***	-0.379^{***}	-0.283***	-0.338***	-0.396***	-0.366***
	(0.097)	(0.098)	(0.098)	(0.099)	(0.109)	(0.103)
Leverage	0.212	0.222	0.249	0.262^{*}	0.426^{**}	0.366^{**}
	(0.171)	(0.145)	(0.183)	(0.136)	(0.187)	(0.181)
$Leverage^2$	0.075	-0.006	0.017	-0.008	-0.039	-0.071
	(0.103)	(0.069)	(0.070)	(0.067)	(0.114)	(0.118)
Stock options	-0.034^{**}	-0.032**	-0.035**	-0.028^{*}	-0.016	-0.011
	(0.015)	(0.015)	(0.015)	(0.015)	(0.019)	(0.020)
Dividends	0.015^{*}	0.032^{***}	0.014	0.027^{***}	0.025^{***}	0.024^{***}
	(0.009)	(0.009)	(0.008)	(0.010)	(0.009)	(0.009)
			Selection	equation		
Net worth	0.680***	0.202**	1.972***	0.298**	-0.031	-0.151***
	(0.257)	(0.102)	(0.580)	(0.133)	(0.034)	(0.044)
Oil price	-0.076	0.003	0.136	0.038	0.060	-0.031
	(0.419)	(0.444)	(0.422)	(0.428)	(0.457)	(0.448)
Oil production	8.258**	7.087^{*}	7.503^{**}	5.631	8.140**	7.512^{*}
	(3.792)	(3.787)	(3.767)	(3.442)	(3.841)	(3.883)
Investments	-0.450	-0.527	-0.778^{*}	-0.514	-1.134^{***}	-1.263^{***}
	(0.372)	(0.385)	(0.435)	(0.339)	(0.426)	(0.425)
Profit diversif.	0.921^{*}	1.283^{**}	0.837^{*}	1.023^{**}	0.605	0.469
	(0.475)	(0.583)	(0.506)	(0.493)	(0.535)	(0.477)
Leverage	3.347^{***}	2.947^{**}	3.793^{***}	2.844^{**}	2.015	2.936^{**}
	(1.102)	(1.223)	(1.228)	(1.133)	(1.274)	(1.147)
$Leverage^2$	-2.717^{***}	-3.266^{***}	-1.814^{**}	-2.557^{***}	-1.764^{**}	-2.120^{**}
	(0.912)	(1.007)	(0.888)	(0.932)	(0.862)	(0.847)
Stock options	0.204^{*}	0.167	0.218^{*}	0.212^{*}	0.172	0.206^{*}
	(0.123)	(0.121)	(0.126)	(0.124)	(0.118)	(0.121)
Dividends	0.178^{*}	0.091	0.217^{**}	0.158	0.155	0.160
	(0.096)	(0.109)	(0.093)	(0.103)	(0.104)	(0.103)

Table 8Hedging choice and derivative strategy

Robust standard error in parentheses. In the selection equation, the dependent variable is equal to 1 when the firm is engaged in risk management activities. In the outcome equation the dependent variable measures the extent of linear notional over total amount of notional hedged. The header of each column indicates the measure of net worth that is used in the estimation, see the text for precise definitions. Average oil is the annual average of the WTI oil prices, Production uncertainty is the coefficient of variation of firm oil production, Oil production is the amount of crude oil produced by the firm in thousands of barrels per day, Investment is defined as firm's capital expenditure over net property, plant, and equipment, Leverage is Total Debt/Assets, Profit diversification is the ratio between revenues from E&P activities and total revenues, Dividends are cash dividends in millions, Stock options is the amount of stock options granted to firms' executives. Missing values for stock options are replaced with zeros. Firms fixed effect are included in the selection equation. *, **, and *** denote significance at, respectively, the 10%, 5% and 1% level. show that when firms' risk spectrum widens to include additional non-hedgeable risk such as production uncertainty, this amplifies the degree of non-linearity in the firms' exposure, hence risk managers should increase their exposure to nonlinear contracts, a result which is empirically found also in Mnasri et al. (2017b) but not in Adam (2009).²⁶Capital expenditures measuring firms' investment propensity are negatively correlated with risk management activities and with the use of linear contracts. Froot et al. (1993) show that firms with large investment programs should be active hedgers, with the amount of nonlinear contracts increasing in the nonlinearity of capital expenditures. Our estimates do not confirm the first prediction, and underscore again the relevant trade-off, driven by collateral constraints, between hedging and investments. On the contrary, we confirm the second prediction, particularly fitting for E&P companies where investment programs are highly nonlinear and strongly dependent on oil prices. We also find that less diversified firms are more likely to engage in financial hedging, consistent with the absence of other forms of diversification via either natural or operational hedging (Kumar and Rabinovitch, 2013). In addition, our results suggest that a lower industrial diversification is also strongly associated with the use of nonlinear contracts. More diversified firms could be more flexible in halting production with adverse oil market conditions. In turn this translates in their resorting more to nonlinear hedging strategies.

Leverage is found to be a significant determinant of the choice to initiate a derivative strategy²⁷ a result consistent with the empirical analysis in Haushalter (2000) and with the theoretical prediction in Smith and Stulz (1985) who are first to emphasize the value of risk management as a factor reducing expected costs of bankruptcy. However, our findings point to a more interesting result, with the lower panel of Table 8 showing a clear non linear relationship: more leveraged producers are increasingly likely to hedge, but when closer to default the sign of this relationship changes. As a possible explanation for this result we refer to the so-called option to default: firms facing very high distress costs avoid hedging and divert money to new projects instead of preserving value for bondholders, because of the limited liability condition. Our findings validate the implications of the risk-shifting model of Fehle and Tsyplakov (2005): firms dynamically adjust their positions with respect

²⁶The result on production uncertainty should be interpreted with some caution as this variable actually resembles more to a sort of firm fixed-effect rather than a time varying measure of production uncertainty. For the same reason we exclude, after testing for their statistical significance, similar regressors from the empirical specification such as price-quantity correlation or the correlation between oil price and firm cash flows, suggested by Froot et al. (1993), Gay et al. (2002), and Brown and Toft (2002) among others.

²⁷Results are qualitatively similar when we substitute leverage with alternative indicators of firms' indebtedness such as the available amount of credit lines.

to both the extent of the hedging activity and the maturity of the hedging position, with the use of risk management displaying a non-monotonic relation with leverage and proximity to financial distress.

In their seminal contributions, Stulz (1984) and Smith and Stulz (1985) characterize hedging in a theoretical framework where managerial risk aversion and incentives are explicitly taken into account. Generally speaking, a risk-averse manager accepts to bear some risk as long as her utility function is convex with respect to the value of the firm; in this case, the manager will be willing to limit the use of hedging and benefit from some volatility in the firm's earning distribution. This could be achieved by designing a convex compensation scheme for the manager, for example by increasing the share of stock options over total executive compensation. In the context of oil and gas companies, Bakke et al. (2016) exploit a change in financial accounting standards as a quasi-natural experiment and find that the decline in managerial risk taking does actually transpose in higher corporate hedging; on the contrary, Croci et al. (2017) do not find empirical support for managerial risk incentives as a driver of the hedging strategy. We proxy the convexity of the managerial compensation scheme via the use of stock options granted to executives; our findings are not able to support the theory as to the choice of entering a derivative contract, while we document a quite robust negative link between the distribution of stock options and the use of linear contracts. We interpret this result as evidence of a managerial preference for strategies that do not cap the upside potential in presence of more convex compensation scheme. Finally, cash dividends exhibit a positive effect on the probability of hedging though the estimates in the selection equation are not always statistically significant. On the contrary, the results in the outcome equation generally indicate an increasing extensive margin achieved through linear contracts for firms with wealthy dividend policies (Adam, 2009).

7 Conclusions

In this paper we used a new hand-collected data set containing detailed information on U.S. oil producers' hedging portfolios. Our study contributes to develop the literature on financial risk management in a sector that has experienced a deep technological (shale revolution) and financial transformation (increasing leverage) over the last decade. Consistently with modern risk management theories, we documented for the first time a strong positive link between net worth and hedging in the oil producing sector. These findings are robust to different model specifications as well as to accounting and market-based measures of net worth. Our results

are of particular interest as they are obtained by analyzing firms where the value of collateral assets strongly depends on the risk factor to be hedged, i.e. the oil price. Exploiting the oil price slumps in 2008 and in 2014-2015 as quasi-natural experiments, we also showed how an oil price shock differently affected risk management practices among E&P firms on the basis of their net worth. Interestingly, in the second oil slump we find a significant role of leverage and credit constraints in reducing the hedging activity. We interpreted this result as a sign of the increasing relevance of the debt component in the firms' "collateral call", a fact that we attribute to the marked increase in leverage following the diffusion of the shale oil technologies. Finally, in the last part of this study we examined more in detail how optimal hedging strategy and the extensive margin of risk management activities interact with firms' net worth. In this case the effect is less apparent though it generally points to a more limited use of linear strategies when firms' net worth increases. The motivations driving this last decision could represent a future interesting research question.

References

- Adam, T. (2009), "Capital expenditures, financial constraints, and the use of options," Journal of Financial Economics, 92, 238–251.
- Adam, T. R. (2002), "Risk management and the credit risk premium," *Journal of Banking* & Finance, 26, 243–269.
- Adler, M. and Detemple, J. B. (1988), "On the Optimal Hedge of a Nontraded Cash Position," Studies in Banking and Finance, 5, 181–197.
- Anderson, S. T., Kellogg, R., and Salant, S. W. (2018), "Hotelling under Pressure," Journal of Political Economy, 126, 984–1026.
- Azar, A. (2017), "Reserve Base Lending and the Outlook for Shale Oil and Gas Finance," Columbia Center on Global Energy Policy working paper.
- Bakke, T.-E., Mahmudi, H., Fernando, C. S., and Salas, J. M. (2016), "The causal effect of option pay on corporate risk management," *Journal of Financial Economics*, 120, 623–643.
- Bjørnland, H. C., Nordvik, F. M., and Rohrer, M. (2017), "Supply Flexibility in the Shale Patch: Evidence from North Dakota," .
- Boyer, M. M. and Filion, D. (2007), "Common and fundamental factors in stock returns of Canadian oil and gas companies," *Energy Economics*, 29, 428 453.
- Brown, G. W. and Toft, K. B. (2002), "How Firms Should Hedge," *The Review of Financial Studies*, 15, 1283–1324.
- Carter, D. A., Rogers, D. A., Simkins, B. J., and Treanor, S. D. (2017), "A review of the literature on commodity risk management," *Journal of Commodity Markets*, 8, 1–17.
- Croci, E., Giudice, A., and Jankensgård, H. (2017), "CEO Age, Risk Incentives, and Hedging Strategy," *Financial Management*, 46, 687–716.
- Dale, S. (2015), "New Economics of Oil," Society of Business Economists Annual Conference, London.
- Domanski, D., Kearns, J., Lombardi, M., and Shin, H. (2015), "Oil and Debt," BIS Quarterly Review March 2015, pp. 55–65.

- Doshi, H., Kumar, P., and Yerramilli, V. (2017), "Uncertainty, Capital Investment, and Risk Management," *Management Science*, 64, 5769–5786.
- Ellwanger, R., Sawatzky, B., and Zmitrowicz, K. (2017), "Factors Behind the 2014 Oil Price Decline," pp. 1–13, Bank of Canada Review.
- Fehle, F. and Tsyplakov, S. (2005), "Dynamic risk management: Theory and evidence," Journal of Financial Economics, 78, 3–47.
- Ferriani, F., Natoli, F., Veronese, G., and Zeni, F. (2018), "Futures risk premia in the era of shale oil," *MPRA working paper*.
- Froot, K. A., Scharfstein, D. S., and Stein, J. C. (1993), "Risk Management: Coordinating Corporate Investment and Financing Policies," *The Journal of Finance*, 48, 1629–1658.
- Gay, G. D., Nam, J., and Turac, M. (2002), "How firms manage risk: the optimal mix of linear and non-linear derivatives," *Journal of Applied Corporate Finance*, 14, 82–93.
- Gilje, E. P. (2016), "Do Firms Engage in Risk-Shifting? Empirical Evidence," Review of Financial Studies, 29, 2925–2954.
- Gilje, E. P. and Taillard, J. P. (2017), "Does Hedging Affect Firm Value? Evidence from a Natural Experiment," *The Review of Financial Studies*, 30, 4083–4132.
- Gilje, E. P., Loutskina, E., and Murphy, D. (2017), "Drilling and Debt," Darden Business School Working Paper No. 2939603.
- Guay, W. and Kothari, S. (2003), "How much do firms hedge with derivatives?" Journal of Financial Economics, 70, 423–461.
- Hamilton, J. D. (2009), "Causes and Consequences of the Oil Shock of 2007-08," Working Paper 15002, National Bureau of Economic Research.
- Haushalter, G., Heron, R. A., and Lie, E. (2002), "Price uncertainty and corporate value," Journal of Corporate Finance, 8, 271–286.
- Haushalter, G. D. (2000), "Financing Policy, Basis Risk, and Corporate Hedging: Evidence from Oil and Gas Producers," *The Journal of Finance*, 55, 107–152.
- Holmström, B. and Tirole, J. (2000), "Liquidity and Risk Management," Journal of Money, Credit and Banking, 32, 295–319.

- Jin, Y. and Jorion, P. (2006), "Firm Value and Hedging: Evidence from U.S. Oil and Gas Producers," *The Journal of Finance*, 61, 893–919.
- Kilian, L. (2014), "Oil Price Shocks: Causes and Consequences," CEPR Discussion Papers.
- Kilian, L. and Hicks, B. (2012), "Did Unexpectedly Strong Economic Growth Cause the Oil Price Shock of 2003-2008?" *Journal of Forecasting*, 32, 385–394.
- Kumar, P. and Rabinovitch, R. (2013), "CEO Entrenchment and Corporate Hedging: Evidence from the Oil and Gas Industry," *Journal of Financial and Quantitative Analysis*, 48, 887–917.
- Lehn, K. and Zhu, P. (2016), "Debt, Investment and Production in the U.S. Oil Industry: An Analysis of the 2014 Oil Price Shock," SSRN working paper.
- Mackay, P. and Moeller, S. B. (2007), "The Value of Corporate Risk Management," *The Journal of Finance*, 62, 1379–1419.
- Mansell, R., Moore, M., Winter, J., and Krzepkowski, M. (2012), "Size, Role and Performance in the Oil and Gas Sector," SPP Research Paper No.12-23.
- Mello, A. S. and Parsons, J. E. (2000), "Hedging and Liquidity," The Review of Financial Studies, 13, 127–153.
- Mnasri, M., Dionne, G., and Gueyie, J.-P. (2017a), "Dynamic corporate risk management: Motivations and real implications," *Journal of Banking & Finance*, 95, 97–111.
- Mnasri, M., Dionne, G., and Gueyie, J.-P. (2017b), "The use of nonlinear hedging strategies by US oil producers: Motivations and implications," *Energy Economics*, pp. 348–364.
- Newell, R. and Prest, B. (2017), "The Unconventional Oil Supply Boom: Aggregate Price Response from Microdata," *NBER working paper*.
- Office of the Comptroller of the Currency (2018), "Comptroller's Handbook: Oil and Gas Exploration and Production Lending,".
- Prest, B. C. (2018), "Explanations for the 2014 oil price decline: Supply or demand?" Energy Economics, 74, 63–75.
- Rampini, A. A. and Viswanathan, S. (2010), "Collateral, Risk Management, and the Distribution of Debt Capacity," *The Journal of Finance*, 65, 2293–2322.

- Rampini, A. A. and Viswanathan, S. (2013), "Collateral and capital structure," Journal of Financial Economics, 109, 466–492.
- Rampini, A. A., Sufi, A., and Viswanathan, S. (2014), "Dynamic risk management," Journal of Financial Economics, 111, 271–296.
- Rampini, A. A., Viswanathan, S., and Vuillemey, G. (2017), "Risk Management in Financial Institutions," Duke University working paper.
- Smith, C. W. and Stulz, R. M. (1985), "The Determinants of Firms' Hedging Policies," The Journal of Financial and Quantitative Analysis, 20, 391–405.
- Stulz, R. M. (1984), "Optimal Hedging Policies," The Journal of Financial and Quantitative Analysis, 19, 127–140.
- Stulz, R. M. (1996), "Rethinking risk management," Journal of Applied Corporate Finance, 9, 8–25.
- Tufano, P. (1996), "Who Manages Risk? An Empirical Examination of Risk Management Practices in the Gold Mining Industry," The Journal of Finance, 51, 1097–1137.

- N. 1188 Raising aspirations and higher education: evidence from the UK's Widening Participation Policy, by Lucia Rizzica (September 2018).
- N. 1189 *Nearly exact Bayesian estimation of non-linear no-arbitrage term structure models*, by Marcello Pericoli and Marco Taboga (September 2018).
- N. 1190 *Granular Sources of the Italian business cycle*, by Nicolò Gnocato and Concetta Rondinelli (September 2018).
- N. 1191 *Debt restructuring with multiple bank relationships*, by Angelo Baglioni, Luca Colombo and Paola Rossi (September 2018).
- N.1192 Exchange rate pass-through into euro area inflation. An estimated structural model, by Lorenzo Burlon, Alessandro Notarpietro and Massimiliano Pisani (September 2018).
- N. 1193 *The effect of grants on university drop-out rates: evidence on the Italian case*, by Francesca Modena, Enrico Rettore and Giulia Martina Tanzi (September 2018).
- N. 1194 *Potential output and microeconomic heterogeneity*, by Davide Fantino (November 2018).
- N. 1195 Immigrants, labor market dynamics and adjustment to shocks in the Euro Area, by Gaetano Basso, Francesco D'Amuri and Giovanni Peri (November 2018).
- N. 1196 Sovereign debt maturity structure and its costs, by Flavia Corneli (November 2018).
- N. 1197 Fiscal policy in the US: a new measure of uncertainty and its recent development, by Alessio Anzuini and Luca Rossi (November 2018).
- N. 1198 Macroeconomics determinants of the correlation between stocks and bonds, by Marcello Pericoli (November 2018).
- N. 1199 Bank capital constraints, lending supply and economic activity, by Antonio M. Conti, Andrea Nobili and Federico M. Signoretti (November 2018).
- N. 1200 *The effectiveness of capital controls*, by Valerio Nispi Landi and Alessandro Schiavone (November 2018).
- N. 1201 Contagion in the CoCos market? A case study of two stress events, by Pierluigi Bologna, Arianna Miglietta and Anatoli Segura (November 2018).
- N. 1202 *Is ECB monetary policy more powerful during expansions?*, by Martina Cecioni (December 2018).
- N. 1203 *Firms' inflation expectations and investment plans*, by Adriana Grasso and Tiziano Ropele (December 2018).
- N. 1204 *Recent trends in economic activity and TFP in Italy with a focus on embodied technical progress*, by Alessandro Mistretta and Francesco Zollino (December 2018).
- N. 1205 Benefits of Gradualism or Costs of Inaction? Monetary Policy in Times of Uncertainty, by Giuseppe Ferrero, Mario Pietrunti and Andrea Tiseno (February 2019).
- N. 1206 Machine learning in the service of policy targeting: the case of public credit guarantees, by Monica Andini, Michela Boldrini, Emanuele Ciani, Guido de Blasio, Alessio D'Ignazio and Andrea Paladini (February 2019).
- N. 1207 Do the ECB's monetary policies benefit Emerging Market Economies? A GVAR analysis on the crisis and post-crisis period, by Andrea Colabella (February 2019).
- N. 1208 The Economic Effects of Big Events: Evidence from the Great Jubilee 2000 in Rome, by Raffaello Bronzini, Sauro Mocetti and Matteo Mongardini (February 2019).
- N. 1209 *The added value of more accurate predictions for school rankings*, by Fritz Schiltz, Paolo Sestito, Tommaso Agasisti and Kristof De Witte (February 2019).

^(*) Requests for copies should be sent to:

Banca d'Italia – Servizio Studi di struttura economica e finanziaria – Divisione Biblioteca e Archivio storico – Via Nazionale, 91 – 00184 Rome – (fax 0039 06 47922059). They are available on the Internet www.bancaditalia.it.

2017

- AABERGE, R., F. BOURGUIGNON, A. BRANDOLINI, F. FERREIRA, J. GORNICK, J. HILLS, M. JÄNTTI, S. JENKINS, J. MICKLEWRIGHT, E. MARLIER, B. NOLAN, T. PIKETTY, W. RADERMACHER, T. SMEEDING, N. STERN, J. STIGLITZ, H. SUTHERLAND, *Tony Atkinson and his legacy*, Review of Income and Wealth, v. 63, 3, pp. 411-444, WP 1138 (September 2017).
- ACCETTURO A., M. BUGAMELLI and A. LAMORGESE, *Law enforcement and political participation: Italy 1861-*65, Journal of Economic Behavior & Organization, v. 140, pp. 224-245, **WP 1124 (July 2017).**
- ADAMOPOULOU A. and G.M. TANZI, *Academic dropout and the great recession*, Journal of Human Capital, V. 11, 1, pp. 35–71, **WP 970** (October 2014).
- ALBERTAZZI U., M. BOTTERO and G. SENE, *Information externalities in the credit market and the spell of credit rationing*, Journal of Financial Intermediation, v. 30, pp. 61–70, WP 980 (November 2014).
- ALESSANDRI P. and H. MUMTAZ, *Financial indicators and density forecasts for US output and inflation*, Review of Economic Dynamics, v. 24, pp. 66-78, **WP 977** (November 2014).
- BARBIERI G., C. ROSSETTI and P. SESTITO, *Teacher motivation and student learning*, Politica economica/Journal of Economic Policy, v. 33, 1, pp.59-72, WP 761 (June 2010).
- BENTIVOGLI C. and M. LITTERIO, Foreign ownership and performance: evidence from a panel of Italian firms, International Journal of the Economics of Business, v. 24, 3, pp. 251-273, WP 1085 (October 2016).
- BRONZINI R. and A. D'IGNAZIO, *Bank internationalisation and firm exports: evidence from matched firmbank data*, Review of International Economics, v. 25, 3, pp. 476-499 WP 1055 (March 2016).
- BRUCHE M. and A. SEGURA, *Debt maturity and the liquidity of secondary debt markets*, Journal of Financial Economics, v. 124, 3, pp. 599-613, WP 1049 (January 2016).
- BURLON L., *Public expenditure distribution, voting, and growth,* Journal of Public Economic Theory,, v. 19, 4, pp. 789–810, **WP 961 (April 2014).**
- BURLON L., A. GERALI, A. NOTARPIETRO and M. PISANI, Macroeconomic effectiveness of non-standard monetary policy and early exit. a model-based evaluation, International Finance, v. 20, 2, pp.155-173, WP 1074 (July 2016).
- BUSETTI F., *Quantile aggregation of density forecasts*, Oxford Bulletin of Economics and Statistics, v. 79, 4, pp. 495-512, **WP 979 (November 2014).**
- CESARONI T. and S. IEZZI, *The predictive content of business survey indicators: evidence from SIGE,* Journal of Business Cycle Research, v.13, 1, pp 75–104, WP 1031 (October 2015).
- CONTI P., D. MARELLA and A. NERI, Statistical matching and uncertainty analysis in combining household income and expenditure data, Statistical Methods & Applications, v. 26, 3, pp 485–505, WP 1018 (July 2015).
- D'AMURI F., *Monitoring and disincentives in containing paid sick leave*, Labour Economics, v. 49, pp. 74-83, **WP 787 (January 2011).**
- D'AMURI F. and J. MARCUCCI, *The predictive power of google searches in forecasting unemployment*, International Journal of Forecasting, v. 33, 4, pp. 801-816, **WP 891 (November 2012).**
- DE BLASIO G. and S. POY, *The impact of local minimum wages on employment: evidence from Italy in the* 1950s, Journal of Regional Science, v. 57, 1, pp. 48-74, **WP 953 (March 2014).**
- DEL GIOVANE P., A. NOBILI and F. M. SIGNORETTI, Assessing the sources of credit supply tightening: was the sovereign debt crisis different from Lehman?, International Journal of Central Banking, v. 13, 2, pp. 197-234, WP 942 (November 2013).
- DEL PRETE S., M. PAGNINI, P. ROSSI and V. VACCA, Lending organization and credit supply during the 2008–2009 crisis, Economic Notes, v. 46, 2, pp. 207–236, WP 1108 (April 2017).
- DELLE MONACHE D. and I. PETRELLA, Adaptive models and heavy tails with an application to inflation forecasting, International Journal of Forecasting, v. 33, 2, pp. 482-501, WP 1052 (March 2016).
- FEDERICO S. and E. TOSTI, *Exporters and importers of services: firm-level evidence on Italy*, The World Economy, v. 40, 10, pp. 2078-2096, **WP 877 (September 2012).**
- GIACOMELLI S. and C. MENON, *Does weak contract enforcement affect firm size? Evidence from the neighbour's court,* Journal of Economic Geography, v. 17, 6, pp. 1251-1282, WP 898 (January 2013).
- LOBERTO M. and C. PERRICONE, *Does trend inflation make a difference?*, Economic Modelling, v. 61, pp. 351–375, **WP 1033 (October 2015).**

- MANCINI A.L., C. MONFARDINI and S. PASQUA, Is a good example the best sermon? Children's imitation of parental reading, Review of Economics of the Household, v. 15, 3, pp 965–993, D No. 958 (April 2014).
- MEEKS R., B. NELSON and P. ALESSANDRI, *Shadow banks and macroeconomic instability*, Journal of Money, Credit and Banking, v. 49, 7, pp. 1483–1516, **WP 939** (November 2013).
- MICUCCI G. and P. ROSSI, *Debt restructuring and the role of banks' organizational structure and lending technologies*, Journal of Financial Services Research, v. 51, 3, pp 339–361, **WP 763 (June 2010)**.
- MOCETTI S., M. PAGNINI and E. SETTE, *Information technology and banking organization*, Journal of Journal of Financial Services Research, v. 51, pp. 313-338, WP 752 (March 2010).
- MOCETTI S. and E. VIVIANO, *Looking behind mortgage delinquencies*, Journal of Banking & Finance, v. 75, pp. 53-63, **WP 999 (January 2015).**
- NOBILI A. and F. ZOLLINO, *A structural model for the housing and credit market in Italy*, Journal of Housing Economics, v. 36, pp. 73-87, **WP 887 (October 2012).**
- PALAZZO F., Search costs and the severity of adverse selection, Research in Economics, v. 71, 1, pp. 171-197, WP 1073 (July 2016).
- PATACCHINI E. and E. RAINONE, Social ties and the demand for financial services, Journal of Financial Services Research, v. 52, 1–2, pp 35–88, WP 1115 (June 2017).
- PATACCHINI E., E. RAINONE and Y. ZENOU, *Heterogeneous peer effects in education*, Journal of Economic Behavior & Organization, v. 134, pp. 190–227, WP 1048 (January 2016).
- SBRANA G., A. SILVESTRINI and F. VENDITTI, *Short-term inflation forecasting: the M.E.T.A. approach*, International Journal of Forecasting, v. 33, 4, pp. 1065-1081, **WP 1016 (June 2015).**
- SEGURA A. and J. SUAREZ, *How excessive is banks' maturity transformation?*, Review of Financial Studies, v. 30, 10, pp. 3538–3580, **WP 1065 (April 2016).**
- VACCA V., An unexpected crisis? Looking at pricing effectiveness of heterogeneous banks, Economic Notes, v. 46, 2, pp. 171–206, WP 814 (July 2011).
- VERGARA CAFFARELI F., One-way flow networks with decreasing returns to linking, Dynamic Games and Applications, v. 7, 2, pp. 323-345, **WP 734** (November 2009).
- ZAGHINI A., A Tale of fragmentation: corporate funding in the euro-area bond market, International Review of Financial Analysis, v. 49, pp. 59-68, WP 1104 (February 2017).

2018

- ADAMOPOULOU A. and E. KAYA, *Young adults living with their parents and the influence of peers*, Oxford Bulletin of Economics and Statistics, v. 80, pp. 689-713, **WP 1038** (November 2015).
- ANDINI M., E. CIANI, G. DE BLASIO, A. D'IGNAZIO and V. SILVESTRINI, *Targeting with machine learning:* an application to a tax rebate program in Italy, Journal of Economic Behavior & Organization, v. 156, pp. 86-102, WP 1158 (December 2017).
- BARONE G., G. DE BLASIO and S. MOCETTI, *The real effects of credit crunch in the great recession: evidence from Italian provinces*, Regional Science and Urban Economics, v. 70, pp. 352-59, **WP 1057** (March 2016).
- BELOTTI F. and G. ILARDI Consistent inference in fixed-effects stochastic frontier models, Journal of Econometrics, v. 202, 2, pp. 161-177, WP 1147 (October 2017).
- BERTON F., S. MOCETTI, A. PRESBITERO and M. RICHIARDI, *Banks, firms, and jobs,* Review of Financial Studies, v.31, 6, pp. 2113-2156, **WP 1097 (February 2017).**
- BOFONDI M., L. CARPINELLI and E. SETTE, *Credit supply during a sovereign debt crisis*, Journal of the European Economic Association, v.16, 3, pp. 696-729, **WP 909** (April 2013).
- BOKAN N., A. GERALI, S. GOMES, P. JACQUINOT and M. PISANI, EAGLE-FLI: a macroeconomic model of banking and financial interdependence in the euro area, Economic Modelling, v. 69, C, pp. 249-280, WP 1064 (April 2016).
- BRILLI Y. and M. TONELLO, Does increasing compulsory education reduce or displace adolescent crime? New evidence from administrative and victimization data, CESifo Economic Studies, v. 64, 1, pp. 15–4, WP 1008 (April 2015).

- BUONO I. and S. FORMAI *The heterogeneous response of domestic sales and exports to bank credit shocks,* Journal of International Economics, v. 113, pp. 55-73, WP 1066 (March 2018).
- BURLON L., A. GERALI, A. NOTARPIETRO and M. PISANI, Non-standard monetary policy, asset prices and macroprudential policy in a monetary union, Journal of International Money and Finance, v. 88, pp. 25-53, WP 1089 (October 2016).
- CARTA F. and M. DE PHLIPPIS, You've Come a long way, baby. Husbands' commuting time and family labour supply, Regional Science and Urban Economics, v. 69, pp. 25-37, WP 1003 (March 2015).
- CARTA F. and L. RIZZICA, *Early kindergarten, maternal labor supply and children's outcomes: evidence from Italy*, Journal of Public Economics, v. 158, pp. 79-102, WP 1030 (October 2015).
- CASIRAGHI M., E. GAIOTTI, L. RODANO and A. SECCHI, A "Reverse Robin Hood"? The distributional implications of non-standard monetary policy for Italian households, Journal of International Money and Finance, v. 85, pp. 215-235, WP 1077 (July 2016).
- CECCHETTI S., F. NATOLI and L. SIGALOTTI, *Tail co-movement in inflation expectations as an indicator of anchoring*, International Journal of Central Banking, v. 14, 1, pp. 35-71, **WP 1025 (July 2015).**
- CIANI E. and C. DEIANA, *No Free lunch, buddy: housing transfers and informal care later in life*, Review of Economics of the Household, v.16, 4, pp. 971-1001, **WP 1117 (June 2017).**
- CIPRIANI M., A. GUARINO, G. GUAZZAROTTI, F. TAGLIATI and S. FISHER, *Informational contagion in the laboratory*, Review of Finance, v. 22, 3, pp. 877-904, **WP 1063 (April 2016).**
- DE BLASIO G, S. DE MITRI, S. D'IGNAZIO, P. FINALDI RUSSO and L. STOPPANI, *Public guarantees to SME borrowing*. A RDD evaluation, Journal of Banking & Finance, v. 96, pp. 73-86, WP 1111 (April 2017).
- GERALI A., A. LOCARNO, A. NOTARPIETRO and M. PISANI, *The sovereign crisis and Italy's potential output*, Journal of Policy Modeling, v. 40, 2, pp. 418-433, **WP 1010** (June 2015).
- LIBERATI D., An estimated DSGE model with search and matching frictions in the credit market, International Journal of Monetary Economics and Finance (IJMEF), v. 11, 6, pp. 567-617, WP 986 (November 2014).
- LINARELLO A., Direct and indirect effects of trade liberalization: evidence from Chile, Journal of Development Economics, v. 134, pp. 160-175, WP 994 (December 2014).
- NUCCI F. and M. RIGGI, *Labor force participation, wage rigidities, and inflation,* Journal of Macroeconomics, v. 55, 3 pp. 274-292, WP 1054 (March 2016).
- RIGON M. and F. ZANETTI, *Optimal monetary policy and fiscal policy interaction in a non_ricardian economy,* International Journal of Central Banking, v. 14 3, pp. 389-436, WP 1155 (December 2017).
- SEGURA A., Why did sponsor banks rescue their SIVs?, Review of Finance, v. 22, 2, pp. 661-697, WP 1100 (February 2017).

2019

CIANI E. and P. FISHER, *Dif-in-dif estimators of multiplicative treatment effects*, Journal of Econometric Methods, v. 8. 1, pp. 1-10, **WP 985** (November 2014).

FORTHCOMING

- ACCETTURO A., W. DI GIACINTO, G. MICUCCI and M. PAGNINI, *Geography, productivity and trade: does* selection explain why some locations are more productive than others?, Journal of Regional Science, **WP 910 (April 2013).**
- ALBANESE G., G. DE BLASIO and P. SESTITO, *Trust, risk and time preferences: evidence from survey data,* International Review of Economics, **WP 911 (April 2013).**
- APRIGLIANO V., G. ARDIZZI and L. MONTEFORTE, Using the payment system data to forecast the economic activity, International Journal of Central Banking, WP 1098 (February 2017).
- ARNAUDO D., G. MICUCCI, M. RIGON and P. ROSSI, *Should I stay or should I go? Firms' mobility across banks in the aftermath of the financial crisis*, Italian Economic Journal / Rivista italiana degli economisti, **WP 1086 (October 2016).**

- BELOTTI F. and G. ILARDI, *Consistent inference in fixed-effects stochastic frontier models*, Journal of Econometrics, WP 1147 (October 2017).
- BUSETTI F. and M. CAIVANO, Low frequency drivers of the real interest rate: empirical evidence for advanced economies, International Finance, WP 1132 (September 2017).
- CHIADES P., L. GRECO, V. MENGOTTO, L. MORETTI and P. VALBONESI, Fiscal consolidation by intergovernmental transfers cuts? Economic Modelling, WP 1076 (July 2016).
- CIANI E., F. DAVID and G. DE BLASIO, *Local responses to labor demand shocks: a re-assessment of the case of Italy*, IMF Economic Review, **WP 1112 (April 2017).**
- COLETTA M., R. DE BONIS and S. PIERMATTEI, Household debt in OECD countries: the role of supply-side and demand-side factors, Social Indicators Research, WP 989 (November 2014).
- CORSELLO F. and V. NISPI LANDI, *Labor market and financial shocks: a time-varying analysis*, Journal of Money, Credit and Banking, **WP 1179 (June 2018).**
- COVA P., P. PAGANO and M. PISANI, Domestic and international macroeconomic effects of the Eurosystem Expanded Asset Purchase Programme, IMF Economic Review, WP 1036 (October 2015).
- D'AMURI F., Monitoring and disincentives in containing paid sick leave, Labour Economics, WP 787 (January 2011).
- D'IGNAZIO A. and C. MENON, *The causal effect of credit Guarantees for SMEs: evidence from Italy*, Scandinavian Journal of Economics, **WP 900 (February 2013).**
- ERCOLANI V. and J. VALLE E AZEVEDO, *How can the government spending multiplier be small at the zero lower bound?*, Macroeconomic Dynamics, **WP 1174 (April 2018).**
- FEDERICO S. and E. TOSTI, *Exporters and importers of services: firm-level evidence on Italy*, The World Economy, **WP 877 (September 2012).**
- GERALI A. and S. NERI, *Natural rates across the Atlantic*, Journal of Macroeconomics, WP 1140 (September 2017).
- GIACOMELLI S. and C. MENON, *Does weak contract enforcement affect firm size? Evidence from the neighbour's court,* Journal of Economic Geography, WP 898 (January 2013).
- GIORDANO C., M. MARINUCCI and A. SILVESTRINI, *The macro determinants of firms' and households' investment: evidence from Italy*, Economic Modelling, WP 1167 (March 2018).
- NATOLI F. and L. SIGALOTTI, *Tail co-movement in inflation expectations as an indicator of anchoring,* International Journal of Central Banking, WP 1025 (July 2015).
- RIGGI M., Capital destruction, jobless recoveries, and the discipline device role of unemployment, Macroeconomic Dynamics, WP 871 (July 2012).
- RIZZICA L., Raising aspirations and higher education. evidence from the UK's widening participation policy, Journal of Labor Economics, WP 1188 (September 2018).
- SEGURA A., Why did sponsor banks rescue their SIVs?, Review of Finance, WP 1100 (February 2017).